Author's response to reviews

Title: Unique Challenges for Appropriate Management of a Teen with Superior Mesenteric Artery Syndrome as a Result of Anorexia Nervosa: A Case Report

Authors:

Philip A Verhoef (pverhoef@mednet.ucla.edu)
Angelika Rampal (arampal@mednet.ucla.edu)

Version: 4 Date: 31 January 2009

Author's response to reviews:

To whom it may concern,

Thank you for your review of our manuscript, “Unique Challenges for Appropriate Management of a Teen with Superior Mesenteric Artery Syndrome as a Result of Anorexia Nervosa: A Case Report.” We greatly appreciate your thoughtful comments regarding the content of our submission. In response to the reviewer’s comments, we have made the following changes:

• We quoted the original reference for the first description of SMA syndrome
• We provided further references where indicated throughout the article
• We explained all abbreviations found in the text
• We abbreviated the case presentation as suggested by the reviewers
• We provided additional information regarding the post-discharge outcome of the patient.
• We omitted Figure 1, as suggested.
• We expanded the discussion to include more references from the literature describing the association between SMA syndrome and anorexia nervosa and the management strategies that have been described.

We also reviewed the formatting of the document to ensure that it conforms to the journal’s style requirements. Please do not hesitate to contact us if you have further questions, clarifications, or requirements in order to facilitate publication.

Sincerely,

Philip A. Verhoef, Ph.D., M.D.
Angelika Rampal, M.D.